Author’s response to reviews

Title: Total saponins of Bolbostemma paniculatum (Maxim.) Franquet exert antitumor activity against MDA-MB-231 human breast cancer cells via inhibiting PI3K/Akt/mTOR pathway

Authors:

Jianwei Dou (djw@mail.xjtu.edu.cn)
Rongguo Shang (shangrongguo@126.com)
Xiaqin Lei (leixxqq@126.com)
Kangle Li (1226299542@qq.com)
Zhanzi Guo (1370831501@qq.com)
Kai Ye (394843905@qq.com)
Xiaojuan Yang (2417375836@qq.com)
Yuwei Li (598405158@qq.com)
Yunyun Zhou (zhouzhouyy@126.com)
Jia Yao (282652404@qq.com)
Qian Huang (410734436@qq.com)

Version: 2 Date: 28 Sep 2019

Author’s response to reviews:

Dear editor:

Thank you so much for pointing the mistakes in our manuscript. We are so sorry for that and revise the paper according to your suggestions.

Because of our carelessness, the author list can not match the authors' contributions. We feel so sorry for the inconvenience, and we correct the mistakes.

Besides, we missed an address of Jian-Wei Dou, Rong-Guo Shang, and Kang-Le Li, so we add an address in the author list. We do not make any changes to the authors, and the order of authors. We have checked the manuscript to upload is a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours, and all relevant figures are clean versions.

Thank you so much!